



## BAISHIDENG PUBLISHING GROUP INC

8226 Regency Drive, Pleasanton, CA 94588, USA

Telephone: +1-925-223-8242 Fax: +1-925-223-8243

E-mail: [bpgoffice@wjgnet.com](mailto:bpgoffice@wjgnet.com) <http://www.wjgnet.com>

---

**Name of Journal:** *World Journal of Clinical Pediatrics*

**ESPS Manuscript NO:** 25596

**Manuscript Type:** Systematic Reviews

### REPLY TO COMMENTS

Dear Editor

As we could not find any specific reviewer comment, we replied only to the Editor comment provided in the edited document (available at the manuscript submission site). Please find below the reply to these comments (required changes have been done in the manuscript and yellow highlighted).

1. The corresponding author address is as per the guideline of the journal
2. The Audio core tip is attached herewith. Please advise any modification (if needed).
3. The references have been updated with the use of PMID and DOI.